Author’s response to reviews

Title: Gender differences in mental health problems among adolescents and the role of social support: results from the Belgian health interview surveys 2008 and 2013

Authors:
Filip Van Droogenbroeck (Filip.Van.Droogenbroeck@vub.be)
Bram Spruyt (Bram.Spruyt@vub.be)
Gil Keppens (Gil.Keppens@vub.be)

Version: 3 Date: 02 Jan 2018

Author’s response to reviews:

Dear editor,

We integrated all you comments.

1. We clarified why consent was not required.

2. We asked a native English speaking colleague to read the declarations and correct any mistakes.

Kind regards,

The authors